Author's response to reviews

Title: Involvement of Toscana virus in cases of meningitis in Tunisia

Authors:

Ons Fezaa (onzfr@yahoo.fr)
Youmna M'ghirbi (youmna.mghirbi@pasteur.rns.tn)
Gianni Gori Savellini (gibbet@libero.it)
Lamia Ammari (lamia_ammariib@yahoo.fr)
Nahed Hogga (nahed.hogga@pasteur.rns.tn)
Henda Triki (henda.triki@pasteur.rns.tn)
Maria Grazia Cusi (cusi@unisi.it)
Ali Bouattour (ali.bouattour@pasteur.rns.tn)

Version: 2 Date: 16 June 2014

Author's response to reviews: see over
Date: 16 June 2014

Editors-in-Chief: Professor Jean-Claude Desenclos

Subject: submission of a Research article for evaluation

Title: Involvement of Toscana virus in cases of meningitis in Tunisia

Keywords: Toscana virus, ELISA, Nested RT-PCR, Viral isolation, Sandflies, Phlebotomus, Tunisia.

Dear Editor,

I, Ali BOUATTOUR, corresponding author of the manuscript, enclosing here with a manuscript entitled «Involvement of Toscana virus in cases of meningitis in Tunisia» to be considered for publication in BMC Infectious Diseases. We choose BMC Infectious Diseases because work deals with viral infectious diseases which is one of the major filed of interest in your journal.

With the submission of this manuscript as a research paper I would like to undertake that:

• All authors of this research paper have directly participated in the planning, execution, or analysis of this study;
• All authors of this paper have read and approved the final version submitted;
• The contents of this manuscript have not been copyrighted or published previously;
• The contents of this manuscript are not now under consideration for publication elsewhere;
• The contents of this manuscript will not be copyrighted, submitted, or published elsewhere, while acceptance by the Journal is under consideration;
• There are no directly related manuscripts or abstracts, published or unpublished, by any authors of this paper;
• My Institute’s (Institute Pasteur of Tunis, TUNISIA) representative is fully aware of this submission.

List of the five potential reviewers:
1- Isabelle Leparc Goffart : isabelle.leparcgoffart@gmail.com
2-Koray Ergünay : ekoray@hacettepe.edu.tr
3-Anna Papa : annap@med.auth.gr
4- Ghram Abdeljelil : abdeljelil.ghram@pasteur.rns.tn
5- Michele Bouloy: michele.bouloy@pasteur.fr

Sincerely yours,

Dr. Ali BOUATTOUR, MVD, PhD
Email: ali.bouattour@pasteur.rns.tn
Institut Pasteur de Tunis, BP74, 1002 Tunis-Bélvédère, TUNISIE
Tel : +216 71 893 340  Fax : +216 71 791 833